Author's response to reviews

Title: Parent and child perceptions of school-based obesity prevention in England: a qualitative study

Authors:

Joanne L Clarke (j.l.clarke@bham.ac.uk)
Tania L Griffin (t.l.griffin.1@bham.ac.uk)
Emma R Lancashire (e.r.lancashire@bham.ac.uk)
Peymane Adab (p.adab@bham.ac.uk)
Jayne M Parry (j.m.parry.1@bham.ac.uk)
Miranda J Pallan (m.j.pallan@bham.ac.uk)

Version: 2 Date: 6 July 2015

Author's response to reviews: see over
Dear Editor

Re: “Parent and child perceptions of school-based obesity prevention in England: a qualitative study”

Please find enclosed a copy of the above original manuscript. As requested, the line numbers have been added and the term Introduction changed to Background. We apologise for these omissions in the original submission.

Thank you for your consideration.

Yours faithfully

Professor Peymane Adab
Professor of Chronic Disease Epidemiology & Public Health